1. Abstract. Please, remove “Eventually, the female was diagnosed with SCAR7” or rephrase it.

This content has been removed from the text

2. Case report.

a. description of birth features should be provided — E.g., GA, head circumference, weight, baby's body length, and reflexes.

The patient had a gestational age of 39 weeks, a head circumference of 34 cm, a birth weight of 2.75 kg and a body length of 49 cm. The rooting reflex, sucking reflex, grip reflex and embrace reflex were normal

b. Could the authors provide a history line of the clinical manifestations of the individual throughout the time?

The patient had normal development from birth to pre-adolescent. She had developed upper and lower limbs tremor 13 years ago and walking unstable, speaking inarticulate and bucking when drunk water and ate fluid food 12 years ago.

c. Why were used MMSE and MoCA instead of other scales specific for individuals with SCA, such as the CCAS scale?

Unfortunately, the CCAS scale has not been formally used in our hospital. After receiving the review opinion, our team immediately translated the CCAS scale into Chinese and tested it on patients. The patient dropped out of school at 13 years of age and could not speak English, so she was unable to complete the PHONEMIC FLUENCY part of the CCAS scale. Nonetheless, this patient had a failure score of 6.

d. Could the authors describe the patient’s vision?

The patient did not have vision problem.